Author's response to reviews

Title: Evaluation of a commercial web-based weight loss and weight loss maintenance program in overweight and obese adults: a randomized controlled trial

Authors:

Clare E Collins (clare.collins@newcastle.edu.au)  
Philip J Morgan (Philip.Morgan@newcastle.edu.au)  
Pennie Jones (PJones@sphealth.com.au)  
Kate Fletcher (Kate.Fletcher@newcastle.edu.au)  
Julia Martin (Julia.Martin@newcastle.edu.au)  
Elroy J Aguiar (Elroy.Aguiar@newcastle.edu.au)  
Ashlee Lucas (Ashlee.Lucas@newcastle.edu.au)  
Melinda Neve (Melinda.Neve@newcastle.edu.au)  
Patrick McElduff (Patrick.Mcelduff@newcastle.edu.au)  
Robin Callister (Robin.Callister@newcastle.edu.au)

Version: 4 Date: 2 November 2010

Author's response to reviews:

Clare Collins  
BSc, Dip N&D, Dip Clin Epi, PhD, advAPD, FDAA  
Professor in Nutrition & Dietetics  
School of Health Sciences, Faculty of Health  
Phone: 61 + 2 49215646  
Fax: 61 + 2 49217053  
Email: Clare.Collins@newcastle.edu.au

To: Editor of BMC Public Health

2nd November, 2010

Dear Editor,

Please find uploaded our revised manuscript MS ID: 9880891104706636, ‘Evaluation of a commercial web-based weight loss and weight loss maintenance program in overweight and obese adults: a randomized controlled trial’.

We have addressed the revisions as follows:-
Formatting changes to be made:
---------------------------------

Major revision

References: please cite all references in numerical order in the text. Reference 14 has not been cited in the correct position.

Author response: This has been correct and reference 14 now becomes reference 23 and the references in between have been adjusted to reflect this new order.

Minor revisions

Citations: please cite the figure as Figure 1 and the table as Table 1 in the text.

Author response: The figure and table have now had the "1" added back in all text and legends.

Sincerely,

On behalf of all authors

Professor Clare Collins

Nutrition & Dietetics, School of Health Sciences

HA12 Hunter Building, University Drive, The University of Newcastle